Author's response to reviews

Title: What differences are detected by superiority trials or ruled out by noninferiority trials? A cross-sectional study on a random sample of two-hundred two-arms parallel group randomized clinical trials.

Authors:

Angèle Gayet-Ageron (angele.gayet-ageron@hcuge.ch)
Thomas Agoritsas (thomas.agoritsas@hcuge.ch)
Christophe Combescure (christophe.combescure@hcuge.ch)
Krisztina Bagamery (Krisztina.Bagamery@finddiagnostics.org)
Delphine S Courvoisier (delphine.courvoisier@hcuge.ch)
Thomas V Perneger (thomas.perneger@hcuge.ch)

Version: 5 Date: 15 October 2010

Author's response to reviews:

We have added as suggested in Authors' contributions: "All authors have read and approved the final manuscript"

We have modified the sentence in the Methods' section, under Statistical analysis related to the Table 3's reference. We changed the sentence with: "if the interaction was not significant, a single P-value for the factor was presented in the last column of the corresponding Table" because to present Table 3 first as a result was non sense.

We have deleted the visible vertical lines from all the tables.

Finally we have carefully read the manuscript and did small changes regarding spaces or modifications from .80 to 80%.